Author's response to reviews

Title: Food patterns reflect more than just eating habits - bias in studies on disease etiology is avoided by appropriate exclusions

Authors:

Agneta Hörnell (agneta.hornell@kost.umu.se)
Anna Winkvist (anna.winkvist@nutrition.gu.se)
Göran Hallmans (goran.hallmans@nutrires.umu.se)
Lars Weinehall (lars.weinehall@epiph.umu.se)
Ingegerd Johansson (ingejerd.johansson@odont.umu.se)

Version: 3 Date: 16 August 2010

Author's response to reviews: see over